Author's response to reviews

Title: Economic hardships in adulthood and mental health in Sweden. The Swedish National Public Health Survey 2009.

Authors:

Johanna Ahnquist (johanna.ahnquist@ki.se)
Sarah P Wamala (sarah.wamala@ki.se)

Version: 2 Date: 12 June 2011

Author's response to reviews:

June 11th 2011

Dear editors,

Re:"Economic hardships in adulthood and mental health in Sweden. The Swedish National Public Health Survey 2009".

We have now, as requested, re-uploaded the document with all the pages in 'portrait' format and also checked that all the text is within the page margins.

Please le us know if we need to make any further changes to the manuscript before you can proceed with peer review.

Kind regards,

Johanna Ahnquist